Author's response to reviews

Title: Using Intervention Mapping for the development of a targeted secure web-based outreach strategy for Chlamydia trachomatis testing in young people at risk.

Authors:

Kevin ATM Theunissen (kevin.theunissen@ggdzl.nl)
Christian JPA Hoebe (christian.hoebe@ggdzl.nl)
Rik Crutzen (rik.crutzen@maastrichtuniversity.nl)
Chakib Kara-Zaïtri (chakib@in-fact.com)
Nanne K de Vries (n.devries@maastrichtuniversity.nl)
Jan EAM van Bergen (JvanBergen@soaids.nl)
Marianne AB van der Sande (marianne.van.der.sande@rivm.nl)
Nicole HTM Dukers-Muijers (nicole.dukers@ggdzl.nl)

Version: 2 Date: 21 December 2012

Author's response to reviews: see over
Dear Editor,

Thank you for giving us the opportunity to revise our manuscript. To give an answer to your comment below we included the following sentence in the Methods section, paragraph 2: "Ethical approval of the Regional Medical Ethics committee in the Netherlands was not necessary, because participants in these interviews were not "subjected to procedures or required to follow certain rules of behavior" [26]."

Editor's comment: Please document in your manuscript whether you received ethics approval for the interviews you have carried out and kindly provide the specific name of the committee that granted this.

Thank you for reconsidering our submission.

Yours sincerely,

On behalf of co-authors Kevin Theunissen, Christian Hoebe, Rik Crutzen, Chakib Kara-Zaïtri, Nanne de Vries, Jan van Bergen and Marianne van der Sande.

Nicole HTM Dukers-Muijrs, PhD, Senior Epidemiologist

Department of Infectious Diseases, South Limburg Public Health Service
P.O. Box 2022, 6160 HA Geleen, The Netherlands,
Telephone: +31(0)45850 6348
Fax: +31(0)455742801
E-mail: Nicole.Dukers@ggdzl.nl